Author’s response to reviews

Title: Public Knowledge of People Visiting Emergency Department Regarding Stroke Symptoms and Risk Factors

Authors:

Elyar Sadeghi-Hokmabadi (aeass@yahoo.com)
Samad Shams Vahdati (sshamsv@yahoo.com)
Reza Rikhtegar (reza_rikhtegar@yahoo.com)
Khazar Ghasempour (Drkhazar.gh@gmail.com)
Aysa Rezabakhsh (aysapharma.rezabakhsh@gmail.com)

Version: 2 Date: 03 Jun 2019

Author’s response to reviews:

Dear editor/editors:

At first, I would like to appreciate for your kind points to rise our manuscript. We also performed any technical or reviewers comments which listed below:

1. Running title (heading) was added.

2. The manuscript revised carefully by expert English editor again and the grammatical errors were corrected as requested and highlighted through the manuscript.

3. Strength/ limitation also was added in the end of the manuscript.